Author’s response to reviews

Title: Descriptive epidemiology of selected birth defects, areas of Lombardy, Italy, 1999

Authors:

Giovanna Tagliabue (biomol@istitutotumori.mi.it)
Paolo Contiero (biomol@istitutotumori.mi.it)
Roberto Tessandori (r.tessandori@asl.sondrio.it)
Fausta Caramaschi (biomol@istitutotumori.mi.it)
Sabrina Fabiano (biomol@istitutotumori.mi.it)
Anna Maghini (biomol@istitutotumori.mi.it)
Andrea Tittarelli (biomol@istitutotumori.mi.it)
Daniele Vergani (biomol@istitutotumori.mi.it)
Maria Bellotti (biomol@istitutotumori.mi.it)
Salvatore Pisani (epidemiol@asl.varese.it)
Maria Letizia Gambino (gaminom@als.varese.it)
Emanuela Frassoldi (biomol@istitutotumori.mi.it)
Enrica Costa (biomol@istitutotumori.mi.it)
Daniela Gada (biomol@istitutotumori.mi.it)
Paolo Crosignani (biomol@istitutotumori.mi.it)

Version: 4 Date: 18 May 2007

Author’s response to reviews:

PHM Editorial Team
Population Health Metrics, Editorial Office

Re Ms. : 8780217601157934
Descriptive epidemiology of selected birth defects, areas of Lombardy, Italy, 1999

Dear Team,
We were pleased to hear that the above paper has been accepted for publication. We have changed the two sentences pointed out by your editorial team to make our meaning even more clear and simple.

We are uploading the paper.

Yours sincerely,

Giovanna Tagliabue and Paolo Contiero